

Updated evidence of the association between toxocariasis and epilepsy: Systematic review and meta-analysis

Jaime Luna, Calogero Edoardo Cicero, Guillaume Rateau, Graziella Quattrocchi, Benoît Marin, Elisa Bruno, François Dalmay, Michel Druet-Cabanac, Alessandra Nicoletti, Pierre-Marie Preux

▶ To cite this version:

Jaime Luna, Calogero Edoardo Cicero, Guillaume Rateau, Graziella Quattrocchi, Benoît Marin, et al.. Updated evidence of the association between toxocariasis and epilepsy: Systematic review and meta-analysis. PLoS Neglected Tropical Diseases, 2018, 12 (7), pp.e0006665. 10.1371/journal.pntd.0006665. hal-01893151

HAL Id: hal-01893151 https://unilim.hal.science/hal-01893151v1

Submitted on 5 Nov 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Citation: Luna J, Cicero CE, Rateau G, Quattrocchi G, Marin B, Bruno E, et al. (2018) Updated evidence of the association between toxocariasis and epilepsy: Systematic review and meta-analysis. PLoS Negl Trop Dis 12(7): e0006665. https://doi. org/10.1371/journal.pntd.0006665

Editor: Agnes Fleury, Instituto de Investigaciones Biomedicas, UNAM /Instituto de Neurologia y Neurocirugía, MEXICO

Received: May 1, 2018

Accepted: July 5, 2018

Published: July 20, 2018

Copyright: © 2018 Luna et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: The authors received no specific funding for this work.

Competing interests: The authors have declared that no competing interests exist.

RESEARCH ARTICLE

Updated evidence of the association between toxocariasis and epilepsy: Systematic review and meta-analysis

Jaime Luna^{1,2®}*, Calogero Edoardo Cicero^{3®}, Guillaume Rateau^{1,2}, Graziella Quattrocchi^{3,4}, Benoit Marin^{1,2,5}, Elisa Bruno³, François Dalmay^{1,2}, Michel Druet-Cabanac^{1,2,5}, Alessandra Nicoletti³, Pierre-Marie Preux^{1,2,5}

1 INSERM, U1094, Tropical Neuroepidemiology, Limoges, France, **2** Univ. Limoges, UMR_S 1094, Tropical Neuroepidemiology, Institute of Neuroepidemiology and Tropical Neurology, CNRS FR 3503 GEIST, Limoges, France, **3** Department of Medical and Surgical Sciences and Advanced Technologies "G.F. Ingrassia", Section of Neurosciences, University of Catania, Catania, Italy, **4** Sobell Department of Motor Neuroscience and Movement Disorders, Institute of Neurology, UCL, London, United Kingdom, **5** CHU Limoges, Centre d'Epidémiologie de Biostatistique et de Méthodologie de la Recherche, Limoges, France

So These authors contributed equally to this work.

* jaimeandreslunam@gmail.com

Abstract

Objective

To gain further insight on the association between human toxocariasis and epilepsy in light of the new evidence in the last years.

Methods

A systematic review was conducted without date and language restriction in the following electronic databases: MEDLINE (PubMed), Ingenta Connect, Science Direct (Elsevier), RefDoc, Scopus, HighWire, Scielo and the database of the Institute of Neuroepidemiology and Tropical Neurology of the Limoges University (IENT). Two investigators independently conducted the search up to November 2017. A pooled odds ratio (OR) was estimated using a random effects model. Meta-regression was conducted to investigate potential sources of heterogeneity.

Results

Database search produced 204 publications. Eleven case-control studies were included that were carried out in 13 countries worldwide. A total number of 4740 subjects were considered (2159 people with epilepsy and 2581 people without epilepsy). The overall pooled OR was 1.69 (95% CI 1.42–2.01) for the association between epilepsy and *Toxocara* spp. seropositivity. A positive association was constantly reported in the restricted analysis (WB as confirmatory or diagnostic test, younger population, and population–based studies). Meta-regression showed no statistically significant association between covariates and outcome.

Conclusion

The updated meta-analysis provides epidemiological evidence of a positive association between *Toxocara* seropositivity and epilepsy. New surveys supported the association, mainly population-based studies. On this basis, health strategies to reduce the impact of *Toxocara* spp are strongly advised. Further research should be performed to understand the physiopathological mechanisms of *toxocara*-associated epileptogenesis.

Author summary

Human toxocariasis is a parasitic zoonosis caused by the larval stages of *Toxocara canis* (*T. canis*), the common roundworm of dogs, and by the roundworm of cats, *Toxocara cati* (*T. cati*). Toxocariasis is one of the most prevalent helminthiasis worldwide. Humans become infected by the ingestion of contaminated food or soil or by direct contact with dogs and cats. Epilepsy is the most common chronic neurological disorder worldwide with a global estimate of about 70 million cases of lifetime epilepsy. Early studies suggested high rate of toxocara exposure (toxocara seropositivity) among people with epilepsy. Despite several studies support the possible role of toxocariasis in the incidence of epilepsy, this association is still debated. We updated our previous research to gain further insight on the association. This study provides new evidence of a positive association between the two diseases. On this basis, health strategies to reduce the impact of *Toxocara* spp are strongly advised. Further studies are necessary to clarify the role of certain parasites in epilepsy.

Introduction

Toxocariasis is a parasitic zoonosis caused by larvae of *Toxocara canis* and *Toxocara cati*, the common roundworms of dogs and cats, respectively.

Toxocariasis is one of the most prevalent helminthiasis worldwide. Even if the parasite tends to be more prevalent in tropical settings where seroprevalence reaches up to 80–90%, in Western countries seroprevalence ranges from 35 to 42% in rural areas and from 2 to 5% in urban areas [1].

Toxocara is a nematode that usually inhabits the small intestine of the host. The female *Toxocara* produces up to 200,000 eggs per day, releasing them to the environment through the feces [2]. Humans become infected by direct contact with dogs or by the ingestion of contaminated food or soil. Ingested eggs develop into juvenile larvae that cross the small intestine and migrate to any organ through the circulatory system, resulting in a multisystemic inflammatory tissue reaction [3]. *Toxocara* larvae migrates to the liver via the portal circulation, then lungs and left heart, from where they disseminate via the systemic circulation, especially to muscles, optic nerves and, in rare cases, the central nervous system [4]. Migrating larvae are attacked by host immune responses, resulting in local inflammation associated with eosinophilia and increased production of cytokines and specific antibodies [5].

Soil is considered as one of the main source of transmission to human beings. The reported prevalence of soil contamination with *Toxocara* spp. eggs is variable between studies, going from a percentage of 6.6 to 87.1% [6]. Infection could also occur as accidental ingestion of embryonated eggs from contaminated water, vegetables, fruit [7], raw or undercooked meat or

organs from paratenic hosts (cows, sheeps or chickens) [8,9]. Additionally, recent studies found a percentage of dogs and cats contaminated with *Toxocara spp.* eggs in their hair that could be a potential risk factor for the transmission of this parasite to other animals and humans [10,11].

The clinical spectrum of toxocariasis in humans varies from asymptomatic infection to severe organ injury [12]. However, the vast majority of infections remain undiagnosed due to the asymptomatic, mild or non-specific clinical nature of infections [13]. Two severe conditions, visceral larva migrans (VLM) and ocular larva migrans (OLM), and two less severe, "covert toxocariasis" and "common toxocariasis" have been described in the literature [14]. *Toxocara* larvae can cross the blood–brain barrier, invading the central nervous system (CNS), leading to neurotoxocariasis [4,15].

Since the first reported case of nematode larvae migration to the brain by Beautyman and Woolf in 1951 [16], more than one hundred cases of neurotoxocariasis have been reported up to date [4,15]. CNS infestation of *T. canis* larvae in humans is thought to be rare, even if in animal models larvae often migrate to the brain [17]. Clinical involvement of the CNS consists of a wide spectrum of neurological manifestations ranging from meningitis, encephalitis, and myelitis, to cerebral vasculitis [4,15].

Early epidemiological reports suggested high seropositivity rates for *T. canis* among people with epilepsy (PWE) [18,19]. Following these preliminary observations several case–controls studies have been carried out suggesting a possible role of toxocariasis in the incidence of epilepsy [6].

An accurate estimate of the association between toxocariasis and epilepsy is needed considering that toxocariasis is one of the most common helminthiasis infections worldwide and it is a potentially preventable disease [7]. For this reason, in 2012 our research teams performed a systematic literature review and a meta-analysis of all the available data [6], finding a positive association between *Toxocara spp*. seropositivity and epilepsy. Despite several studies support the possible role of toxocariasis in the incidence of epilepsy, this association is still debated [1]. Considering that new studies have been performed, we updated our previous research to gain further insight on the association between human toxocariasis and epilepsy and to evaluate any methodological improvement in the field.

Methods

Search strategy

We updated our previous literature search [6]. The aim was to identify any new published and unpublished evidence on the association between toxocariasis and epilepsy. The new systematic search has been independently conducted by two investigators (JL and CEC) in the following electronic databases: MEDLINE (PubMed), Ingenta Connect, Science Direct (Elsevier), RefDoc, Scopus, HighWire, Scielo and the database of the Institute of Neuroepidemiology and Tropical Neurology of the Limoges University (IENT).

To search for the association between epilepsy and toxocariasis the following research strings and Boolean operators have been entered in each database: "epilepsy" AND "toxo-cariasis" AND "epidemiology". In order to find all the available articles, the search term "toxo-carosis" has also been used. The research has been performed without date or language restriction. Titles and abstract have been screened to select relevant studies, which have been fully read. For all the selected articles references have been searched for other relevant articles. When necessary, corresponding authors were contacted. Experts in the field were also contacted to find out other eventual non-published studies. The search was realized up to November 2017.

Study selection

Considering epilepsy as the outcome and toxocariasis as the exposure, studies had to fulfill the following inclusion criteria, already used in the previous meta-analysis [6], to be included:

- 1. Presence of a control group (people without epilepsy, PWOE);
- 2. Information about methods used to assess epilepsy;
- 3. Serological or histopathological detection of toxocariasis;
- 4. Information about methods and criteria used for case finding and control selection;
- 5. Possibility to determine the sample size of each of the following four groups in aggregated data: people with epilepsy seropositive for toxocariasis (PWE *Toxocara* spp.+), people with epilepsy not seropositive for toxocariasis (PWE *Toxocara* spp.-), people without epilepsy seropositive for toxocariasis (PWOE *Toxocara* spp.+), people without epilepsy not seropositive for toxocariasis (PWOE *Toxocara* spp.+), people without epilepsy not seropositive for toxocariasis (PWOE *Toxocara* spp.+), people without epilepsy not seropositive for toxocariasis (PWOE *Toxocara* spp.+), people without epilepsy not seropositive for toxocariasis (PWOE *Toxocara* spp.+), people without epilepsy not seropositive for toxocariasis (PWOE *Toxocara* spp.+), people without epilepsy not seropositive for toxocariasis (PWOE *Toxocara* spp.+), people without epilepsy not seropositive for toxocariasis (PWOE *Toxocara* spp.+), people without epilepsy not seropositive for toxocariasis (PWOE *Toxocara* spp.-).

Systematic reviews, case reports or commentaries were not included in the study selection. Surveys including only acute symptomatic seizures or specific seizure patterns or epileptic syndromes were excluded.

Data extraction

Two investigators (JL and CEC) independently examined the identified articles. Discrepancies were discussed and agreement was reached for all the articles included in the analysis. For each survey the following information have been entered in an *ad hoc* created database: author, country, study design, study population (number, age groups, and gender), study setting, recruitment methods, epilepsy definition and confirmation methods, and exposure assessment.

Association between epilepsy and toxocariasis has been calculated for each included survey giving a crude odds ratio (OR) and their relative 95% confidence interval (CI). Furthermore, *a priori* and *a posteriori* statistical power have been calculated. *A priori* statistical powers were calculated following the hypothesis that the objective of the survey was to identify a minimum OR of 2 with one control per case, based on the number of PWE and the percentage of *Toxocara* spp. seropositivity in PWOE. *A posteriori* statistical power were calculated upon the results of the surveys. In both cases a 5% alpha risk was considered. Powers were calculated using Epi-Info statistical software 6.4. [20]

Meta-analysis

To estimate the association between epilepsy and toxocariasis a meta-analysis has been performed using a random effects model [21]. The Cochran Q test and I^2 were used to examine statistical heterogeneity in the meta-analysis [22]. Further restricted analyses have been conducted: i) We applied the analysis only to the studies using Western Blot (WB) as diagnostic or confirmatory test for toxocariasis [23–27], considering that WB is as sensitive but more specific than enzyme-linked immunosorbent assay (ELISA) [28]. ii) We performed a restricted analysis of studies including young population (<18 years) [29–31], in order to account for the different age groups considered. iii) We also conducted a restricted analysis of the studies performed in general population settings [23,27,32], taking into consideration that population-based case–control studies can avoid selection biases compared with hospital-based studies [33]. The publication bias was evaluated through the funnel plot and the Egger's test was used for detecting asymmetry. Meta-regression was conducted to investigate potential sources of heterogeneity. We used random effects meta-regression to examine whether any covariate was associated with the outcome. Several study level co-variables were considered. For each meta-regression, the p value and the tau² were reported for each covariate.

The meta-analysis and meta-regression were performed using the statistical software Stata v11.1 (Stata Corporation, College Station, TX, USA). The PRISMA (Preferred Reporting Items for Systematic reviews and Meta-Analyses) statement [34] was followed as a guide to report this study (PRISMA checklist—S1 File).

Results

Literature search

The detailed steps of the updated literature research and selection process are shown in the Fig 1. A PRISMA flowchart is also shown (S1 Fig). Database search produced 204 publications. After title and abstract examination 29 articles full text were selected. Searching through references produced additional two publications [31,35]. The removal of duplicates permitted to identify 15 documents. Full text review of the documents allowed us to exclude four of them for not fulfilling the inclusion criteria. In particular, one publication was excluded since there was no control group [19]. Another two publications were also excluded because they provided insufficient information about the methodology used in the surveys [36,37]. We found two new studies from the same research group, and we excluded the first one from the analysis [38] because the same population was considered [32]. Nevertheless, the materials and methods of this last study had been previously detailed in the first publication [38], therefore we assessed the methodological aspects considering both publications. Briefly, 11 studies were considered for the analysis.

Included studies

Eleven case-control studies were included, seven from our previous systematic review [23–26,29,30,39] and four new ones [27,31,32,35]. A total number of 4740 subjects were considered, 2873 additional subjects compared to our previous meta-analysis (1867 subjects) [6]. This updated study considers more than twice as many cases and controls (2159 PWE and 2581 PWOE) compared to our previous study (850 PWE and 1017 PWOE) [6].

The studies were carried out in 13 countries: one in North America [USA] [29], one in South America [Bolivia] [23], two in Europe [Italy] [25,30], three in Asia [India [27], Iran [35], Turkey [39]] and four in Africa [Burundi [24], Egypt [31], Tanzania [26], one of them included 5 African countries: Ghana, Kenya, South Africa, Tanzania, Uganda [32]] (Fig 2). Four studies were performed in rural [23,24,26,32] and seven in urban settings [25,27,29–31,35,39].

The large majority of the surveys were hospital-based studies [24-26,29-31,35,39] and only three population-based case-control studies were found [23,27,32]. Three studies were conducted on a younger population: two considered a population aged 1–17 years [29,30] and one included patients aged 14 years or younger [31]. Seven studies considered no age restriction [23-25,27,32,35,39] while another one excluded children aged 10 years or younger [26]. The general characteristics of the included studies are shown in Table 1.

Four studies [23–25,27] used the epilepsy definition proposed by the International League Against Epilepsy (ILAE) in 1993 [40]. One survey [32] adopted the definition given by *Edwards et al.* in 2008 [41], based on the ILAE active epilepsy definition. Three studies adopted specifics definitions: *Glickman et al.* [29] considered the definition proposed by *Alter* [42] in 1972. *Arpino et al.* [30] considered a general definition of "positive seizure history" as cases entry criteria and *Winkler et al.* [26] defined epilepsy according to the World Health Organization (WHO)



https://doi.org/10.1371/journal.pntd.0006665.g001



A) Bolivia, B) Burundi, C) Egypt, D) Ghana, E) India, F) Iran, G) Italy, H) Kenya, I) South Africa, J) Tanzania, K) Turkey, L) Uganda, M) USA. Fig 2. Countries of the studies included on the association between toxocariasis and epilepsy.

https://doi.org/10.1371/journal.pntd.0006665.g002

Neurosciences Research Protocol proposal [43]. Cryptogenic epilepsy was investigated in three surveys using the ILAE definition [31,35,39].

Considering the type of seizures, five studies [23–25,35,39] applied the classification of epilepsies and epileptic syndromes proposed by the ILAE in 1981 [44], while only one [26] used an adjusted classification for rural African hospitals suggested in 2007 [45]. A standard electroencephalography (EEG) recording was performed in five studies [23–25,27,30] in order to determine the accuracy of seizures classification in patients with a clinical diagnosis of epilepsy.

A neurologist, or an epileptologist, confirmed the epilepsy diagnosis in six studies through anamnesis and complete neurological examination [23-27,32]. All PWE were prevalent cases and only three of the studies clearly specified the inclusion of active epilepsy cases [24,27,32].

Controls have been recruited from the general population in the population-based surveys [23,27,32]. Controls were selected from the same geographical area than cases (rural community or province), and close relatives were excluded by choosing people from different households [23]. In the hospital-based studies, controls were outpatients [30] or in-patients attending the same hospital of cases [29] or people going to hospital for vaccination or blood tests [24,25] or volunteers or relatives [26,31,35,39]. A negative history for seizures [24–27,30,31,35,39] and for both seizures and other neurological diseases [24,25,27,30,35] was the inclusion criteria for controls. Six surveys had a matched case-control design [23–25,27,31,32], age was the most common matching criteria.

Sociodemographic variables and possible risk factors were collected in eight studies [24,25,27,29–32,39] using a questionnaire administered to cases and control subjects.

Several technical approaches were used to assess the presence of anti-*Toxocara* spp. anti-bodies in sera as antibodies-ELISA (Ab-ELISA) commercial or in-house kits [29–32,39], or

				PWE Ascertainment		PW	Exposure	
References	Country	Study design	Sources	Epilepsy definition and classification	Confirmation	Sources	Matching Criteria	Examinations
Glickman et al., J Pediatr 1979 [29]	USA	Case- control	Pediatric Hospital	Alter et al., 1972 Cases known by hospital		Outpatients or hospitalized controls	None	Sera Ab-ELISA
Arpino et al., Epilepsia 1990 [<u>30]</u>	Italy	Case- control	Pediatric Hospital	Not specified Cases known Pediatric Hospi ("positive seizure by hospital history")		Pediatric Hospital	None	Sera Ab-ELISA
Nicoletti et al., Neurology 2002 [<u>23</u>]	Bolivia	Case- control	General population	ILAE 1993; ILAE Neurologist General 1 1981		General population	Sex, age ± 5 years, same community	Sera Ab- ELISA, WB
Akyol et al., Seizure 2007 [39]	Turkey	Case- control	Hospital, consecutively enrolled	Not specified ("idiopathic epilepsy"); ILAE 1981	Cases known by hospital	nown Volunteers, source None spital not specified		Sera Ab-ELISA
Nicoletti et al., Epilepsia 2007 [24]	Burundi	Case- control	PWE identified by local health centers	ILAE 1993; ILAE 1981	Neurologist	Controls coming to hospital for vaccination or PWE neighbors	Age ± 5 years, no blood relationship, same province	Sera WB
Nicoletti et al., Epilepsia 2008 [25]	Italy	Case- control	Epilepsy center, randomly selected	ILAE 1993; ILAE 1981	Neurologist	Subjects who went to hospital for hematological check, consecutively enrolled		Sera WB
Winkler et al., Trans R Soc Trop Med Hyg. 2008 [26]	Tanzania	Case- control	Hospital, age > 10 years	WHO (1993), Winkler et al., 2007	Neurologist	Relatives and volunteers	Relatives and None volunteers	
Singh et al., Epilepsia 2012 [27]	India	Case- control	General population	ILAE 1993	Epileptologist, EEG, MRI	General population	Sex, age \pm 2 years for age > 10, age \pm 1 year for age \leq 10, same area	Sera Ab-ELISA, WB
El-Tantawy et al., Ajidm 2013 [31]	Egypt	Case- control	Pediatric Hospital	Cryptogenic epilepsy; ILAE Commission on Classification and Terminology, 2005– 2009	Cases known by hospital	Volunteers outpatients with no personal or family history of seizures	Volunteers outpatients with no personal or family history of seizuresAge: three age groups (≤ 5 years, $>5 to \leq 10$ years, $> 10 to \leq 14$ years)	
Zibaei et al., Hindawi 2013 [<u>35]</u>	Iran	Case- control	Hospital	Not specified ("idiopathic epilepsy") ILAE 1981	Cases known by hospital	Volunteers (healthcare workers or patients' relatives)	None	Sera Ab-ELISA, WB
Kamuyu et al., PLOS NTD 2014 [32]	South Africa, Tanzania, Uganda, Kenya, Ghana	Case- control	General population	Edwards et al. 2008, ILAE 1993	Neurologist	General population	Group-matched by age, same country	Sera Ab-ELISA

Table 1. Description of the studies included looking for an association between toxocariasis and epilepsy.

Ab-ELISA, antibodies enzyme-linked immunosorbent assay; CSF, cerebrospinal fluid; ILAE, International League Against Epilepsy; PWE, people with epilepsy; PWOE, people without epilepsy; WB, Western Blot; WHO, World Health Organization.

https://doi.org/10.1371/journal.pntd.0006665.t001

Western Blot (WB) [24,25] or Ab-ELISA screening followed by WB confirmation [23,26,27,35]. Laboratories performing the analysis were blind to the case-control status of sera samples in three studies [23–25].

Association between toxocariasis and epilepsy

Statistical significant association between *Toxocara* spp. seropositivity and epilepsy was found in five surveys [23,25,29,30,32]. Significant crude ORs ranged from 1.58 to 2.85. Borderline

References	PWE (n)	PWOE (n)	Seropositivity in PWE (n)	Seropositivity in PWOE (n)	Seropositivity in PWE (%)	Seropositivity in PWOE (%)	A priori statistical power (%)*	A posteriori statistical power (%)	OR (95% CI)	p-value
Glickman et al., J Pediatr 1979 [29]	84	108	19	11	22.6	10.2	33.6	65.3	2.58 (1.15– 5.77)	0.018
Arpino et al., Epilepsia 1990 [30]	91	214	20	26	21.9	12.1	40.1	59.3	2.04 (1.07- 3.88)	0.028
Nicoletti et al., Neurology 2002 [23]	113	233	28	28	24.8	12.0	47.6	85.0	2.41 (1.35- 4.31)	0.002
Akyol et al., Seizure 2007 [<u>39</u>]	100	50	12	4	12.0	8.0	32.8	11.3	1.57 (0.48– 5.14)	0.454
Nicoletti et al., Epilepsia 2007 [24]	191	191	114	97	59.7	50.8	90.9	41.0	1.43 (0.96– 2.15)	0.080
Nicoletti et al., Epilepsia 2008 [25]	231	201	38	13	16.4	6.5	55.8	89.6	2.85 (1.47– 5.51)	0.001
Winkler et al., Trans R Soc Trop Med Hyg. 2008 [26]	40	20	19	8	47.5	40.0	33.5	8.0	1.36 (0.46– 4.03)	0.582
Singh et al., Epilepsia 2012 [27]	106	106	5	6	4.7	5.7	27.0	4.9	0.83 (0.24– 2.79)	0.757
El-Tantawy et al., Ajidm 2013 [<u>31</u>]	132	60	64	28	48.5	46.7	79.2	4.3	1.08 (0.58– 1.98)	0.815
Zibaei et al., Hindawi 2013 [35]	85	85	10	3	11.8	3.5	16.0	52.0	3.64 (0.97– 13.75)	0.043
Kamuyu et al., PLOSntd 2014 [32]	986	1313	308	293	31.2	22.3	100.0	99.8	1.58 (1.31– 1.91)	<0.001

Table 2. Results of the included case-control studies on the association between toxocariasis and epilepsy.

*Statistical power assuming an odds ratio equal to 2 with a type I error equal to 5% and one control per case.

OR, crude odds ratio; PWE, people with epilepsy as cases; PWOE, people without epilepsy as controls; seropositivity, presence of antibodies anti-Toxocara canis; 95% CI, 95% confidence interval.

https://doi.org/10.1371/journal.pntd.0006665.t002

statistical significance was found in two studies [24,35]. *A posteriori* statistical power was higher than 80% in only three studies [23,25,32]. *A priori* and *a posteriori* statistical power and crude ORs of the included studies are shown in Table 2.

Meta-analysis

A meta-analysis was performed considering the 11 studies included. The overall pooled OR was 1.69 (95% CI 1.42–2.01) for the association between epilepsy and *Toxocara* spp. seropositivity (Fig 3). The test of heterogeneity was not significant (p = 0.348) and the I² was 10.1%.

Five studies used Western Blot as diagnostic or confirmatory test [23-27,35], we excluded one of them in the restricted analysis because no controls were seropositive using WB [35]. A common OR of 1.79 (95% CI 1.24–2.59) was found [23-27] (Fig 4A). When we restricted the meta-analysis to the three studies considering a young population [29–31], we found a



Fig 3. Meta-analysis of studies on the association between toxocariasis and epilepsy.

https://doi.org/10.1371/journal.pntd.0006665.g003

common OR of 1.71 (95% CI 1.02–2.87) (Fig 4B). The meta-analysis was at last restricted to the three population-based studies [23,27,32] leading to an OR of 1.68 (95% CI 1.17–2.40) (Fig 4C). The heterogeneity tests were not significant in the three restricted analyses.

To assess the publication bias, the funnel plot is shown in the <u>S2 Fig</u> and non-significant Egger's test was found (p = 0.454).

Meta-regression

In order to explore the sources of heterogeneity, the following study level co-variables were considered: time period (tau² = 0.00, p = 0.132), continent (tau² = 0.00, p = 0.181), technique to assess *Toxocara* exposure (tau² = 0.015 p = 0.718), age group population (tau² = 0.015, p = 0.937), case and control ascertainment source (tau² = 0.029, p = 0.813), epilepsy diagnosis confirmation by a neurologist (tau² = 0.011, p = 0.818) and epilepsy definition used (tau² = 0.019, p = 0.809). Meta-regression showed no statistically significant association between covariates and outcome.

Discussion

Today, the association between toxocariasis and epilepsy is still debated which is related to different results found in the literature and the lack of understanding of physiopathological mechanisms between toxocariasis and epileptogenesis. This updated meta-analysis provides epidemiological evidence of the association between *Toxocara* spp seropositivity and epilepsy, with a pooled overall OR of 1.69 (95% CI 1.42–2.01). The association is consistent but slightly lower than the previous published meta-analysis (OR 1.92, 95% CI 1.50–2.44) [6]. A positive

A) Western Blot as diagnostic or confirmatory test



B) Young population



C) Population-based studies



Fig 4. Meta-analysis of studies on the association between toxocariasis and epilepsy: a) Western Blot as diagnostic or confirmatory test, b) young population, c) population-based studies.

https://doi.org/10.1371/journal.pntd.0006665.g004

association was constantly reported in the restricted analysis (WB as confirmatory and diagnosis, younger population, and population–based studies).

New evidence

We updated our previous meta-analysis [6] to gain further insight in the association in light of the new evidence in the last years. We applied similar criteria for the selection of relevant papers in order to include studies with an adequate methodological approach to assess the association.

The four new articles included explicitly investigated the association between epilepsy and toxocara exposition using a case-control design. Case recruitment was performed in a general population setting in two studies [27,32], adopting a two-stage and three-stage design for the detection of cases. For the other two studies, cases ascertainment was performed in a hospital-based setting [31,35]. Controls were selected among the general population of the same communities where the population-based studies were set. In the hospital-based studies, controls have been recruited from the outpatient services of the same hospitals where the cases have been recruited or even from patients' relatives without epilepsy [35]. A major methodological difficulty with case-control studies is the selection of appropriate control individuals [46]. Selection of population controls ensures that the distribution of exposures in the controls can be readily extrapolated to the population [47]. This is a main point when investigating an association with public health implications. Furthermore, the two population-based studies [27,32] estimated the sample size calculation with satisfactory statistical power (80%).

The new evidence allowed us: i) to consider more than twice as many cases and controls compared to our previous study improving the statistical power to estimate more accurately the pooled OR of the association, ii) to include eight additional countries worldwide which strengthens the consistency criterion for causality, iii) to conduct restricted analysis considering population-based case-control studies.

Population-based versus hospital-based studies

Overall, the majority of the studies included were performed in a hospital setting. Hospital based studies are frequently prone to selection bias, as the cases being followed in the hospital probably present more severe form of epilepsy compared to the general population. Children suffering from severe encephalopathies, or patients with a history of tonico-clonic generalized seizures more frequently need specialized care in a hospital setting, while subjects presenting milder symptomatology or focal seizures without secondary generalization, might be underrepresented. The difference is even more striking in low and middle-income countries, where the elevated costs of access to specialized care further reduce the representation of milder forms. This bias may partially account for the lack of association found in some of the hospital-based studies, because Toxocara spp seropositivity has been showed to be associated with focal seizures (adjusted OR 4.69; 95% CI 2.24–9.80) [25]. Furthermore, controls selected in a hospital-based setting tend to share similar characteristics of the cases, while controls taken from the community better represent the general population. The inclusion of volunteers as controls could be another source of bias since volunteers are frequently healthier than the general population. In order to take into account these sources of bias, we have restricted our analysis to the 3 population-based studies [23,27,32], finding a common OR of 1.68 (95% CI 1.172.40), a result close to the one obtained when considering all the studies thus minimizing the impact of the aforementioned selection bias. One of the new population-based studies was a large survey performed in five African nations with an unprecedented sample size and very good *a-priori* and *a posteriori* statistical power [32]. Toxocariasis seropositivity was assessed using Ab-ELISA in sera and it was reported a variable proportion of cases and controls exposure to multiple infections. However, the relative excess risk due to interaction was estimated and adjusted to consider the effect of multiple parasitic infections.

Exposure to Toxocara in young populations

To take in account the possible effect of age on the exposure to *Toxocara* spp, the analysis has been restricted to the studies investigating a young population (<18 years), giving a common OR of 1.71 (95% CI 1.02–2.87), a result now borderline significant when compared to the previous meta-analysis of 2.23 (95% CI 1.36-3-69) [6]. The updated analysis on young population included only one new study published in the last six years, conducted in Egypt by El Tantawy (2013) which found no difference in *Toxocara* spp antibodies rate between cases and controls [31]. Nevertheless, there are some methodological issues to consider as the higher prevalence of subjects with a rural background in the controls compared to the cases. This could raise the number of Toxocara spp positive subjects in the control group as living in rural setting increases the possibility of being exposed to *T. canis* eggs, especially in a country with a high soil contamination rate such as Egypt [48]. Moreover, the calculated *a posteriori* statistical power was 4.3%, which increases the probability of making a type II error. Age has also been used as matching criteria in all the included studies because it seems to contribute to Toxocara spp. exposure. Generally the chance of being exposed to an infectious agent increases with age, however it has also been demonstrated that younger age is associated with higher Toxocara spp seropositivity [7] probably due to habits that increase the contact with *Toxocara* spp eggs infected soil (such as playgrounds). Interestingly, toxocariasis is more common in childhood, while neurotoxocariasis cases are observed more frequently in adults [4].

Epilepsy and types of seizures

Another factor that affects the interpretation of the association between epilepsy and toxocariasis depends on the lack of a clear classification of the etiologies of epilepsies and the different types of seizures. It should be expected, according to the hypothesized physiopathology, that *Toxocara* spp should be associated with focal seizures, with or without secondary generalization rather than generalized seizures. However only few studies provided a stratified analysis according the type of seizures. This association has been confirmed in two of the studies included [23,25], while *Zibaei et al.* (2013) found a borderline significant association in a population comprised largely by patients with partial seizures (82.4%) [35]. Furthermore, seizure have been classified just on a clinical basis in the majority of studies, and accuracy of seizures classification when performed just on clinical ground is questionable. Electroencephalography (EEG) can help differentiating the type of seizures but it is rarely available in epidemiological studies, above all in low and middle-income countries.

Techniques to evaluate toxocariasis seropositivity

Another possible limitation is due to the heterogeneity of techniques (ELISA, WB, or both) and kits (commercial or in house) used in the different studies to evaluate toxocariasis seropositivity. The standard serological test for the diagnosis of toxocariasis is an ELISA based on secretory-excretory antigens (TES) from *Toxocara canis* larvae of the second stage [49] and it is the most commonly used in epidemiological surveys. ELISA for the detection of specific IgG antibodies to TES in serum has a sensitivity of 78% and a specificity of 92% for the diagnosis of VLM, although cross-reactions with other nematode infections (e.g. *Ascaris lumbricoides*, *Stronglyloides*), reduce its specificity, particularly in tropical areas [50]. The use of fractionated native TES in the western blot (WB) assay overcomes the issues with cross-reactions to other nematodes in ELISA assays [28] thus screening with the indirect TES–IgG–ELISA, followed by confirmation with the TES–WB, is considered an effective approach [5]. Therefore, we have chosen to perform a restricted analysis of those studies that employed Western Blot as confirmatory test. We found a common OR of 1.79 (95% CI 1.24–2.59) that is close to the common OR obtained when considering all the studies.

Statistical power of the studies

A possible explanation for the lack of association on some of the included studies may rely upon the low statistical power that these studies possessed to find a significant association between epilepsy and toxocariasis. In fact, only three studies had *a posteriori* statistical power over 80%, a value usually considered as appropriate when planning research studies, and they all found the association to be significant [23,25,32]. Nevertheless, studies that had a statistical power between 50% and 80% also found a significant association [29,30,35], while the other studies, which statistical powers ranged from 4.3% to 41.0% did not find any association [24,26,27,31,39].

Causality

A major limitation in the interpretation of the association is due to the fact that all the studies included in our meta-analysis have been conducted on prevalent epilepsy cases, thus considering a cross-sectional approach at the same time of both the exposure (anti-Toxocara spp antibodies) and the outcome (epilepsy). This does not allow us to exclude a reverse causality in which the development of epilepsy acts as a risk factor for the infection with Toxocara spp. In fact, epilepsy itself is a risk factor for frequent falls to the ground that increase the chance of Toxocara spp infection especially in areas with high levels of soil contamination. Children with epilepsy and mental retardation, such as those suffering from severe perinatal encephalopathies, may have abnormal behaviors (e.g. pica) that increase the chance of Toxocara spp infection [29]. To clarify the temporal relation, it is mandatory to investigate the presence of anti-Toxocara spp antibodies in a population of incident patients with epilepsy, however no case control study considering incident PWE has been performed to assess the association. A prospective population-based survey including new cases will provide the most reliable epidemiological evidence of association but this does not necessarily imply causation [14]. Therefore, it has been proposed by Wagner and Newton (2009) that causal relationship can be established only by measuring the reduction in the incidence of epilepsy following the eradication of the putative helminth (Toxocara spp) [51].

Pathogenic mechanisms: Toxocariasis and epilepsy

The role of exposition to *Toxocara spp* and the development of epilepsy is still a matter of debate. While it has been demonstrated that the parasite can migrate to the CNS and cause a variety of clinical manifestation [14], there are no sufficient evidence supporting a direct epileptogenic role. Some pathogenic mechanisms have been proposed to explain the possible implication of Toxocariasis in epileptogenisis. First, studies have shown that *Toxocara* larvae could induce a granulamotous reaction. An acute granulomatous reaction may cause acute symptomatic seizures, and may leave fibrous scars after resolution and chronic granulomatous lesions maybe could lead to epilepsy [51]. Granulomas has been found in the cortical or sub-

cortical brain areas through magnetic resonance imaging (MRI) in neurotoxocariasis cases [15]. Second, animal models have showed that the presence of *Toxocara* spp larvae in the brain is responsible for an increase in the permeability of the blood brain barrier, and the production of proinflammatory cytokines leading to neuronal damage [1]. Similar physiopathological mechanisms have been hypothesized to play a major role in the epileptogenesis due to neuro-cysticercosis [52] suggesting that the two parasites may share a common epileptogenetic pathway. Third, there is growing evidence that autoantibodies against neuronal elements may play a role in some type of epilepsies [53,54]. Clues of the autoimmune nature of epilepsy came from the presence of antibodies to a major excitatory neuro-transmitter in the CNS [55,56]. Toxocariasis produces a marked immune response including autoantibodies production [57], however it is unclear the relation of these immune mechanisms in epileptogenesis. Further research is needed to clarify the potential implication of these pathways in epilepsy.

Strengths and limitations

This work relies on certain strengths. First, the broad literature search in eight large databases without date and language restriction, which allow us to screen the available literature in the subject. Second, international guidelines on meta-analysis of observational studies were followed. Third, this update included studies carried out in rural and urban settings in different geographic areas worldwide. Lastly, no apparent publication bias exists in the studies included in the meta-analysis. However, we recognize that the funnel plots should be interpreted with caution when the studies are few.

Some limitations could be discussed. First, only few new studies were performed to evaluate the association, however, this update considers more than twice as many cases and controls compared to our previous study [6]. Second, the statistical power in most of the surveys was well below the minimum of 80% that is considered appropriate. Third, the use of definitions, methodologies and techniques were heterogeneous among the studies. Fourth, certain potential risk factors (pica or exposure to pets) were not systematically assessed. Fifth, relevant clinical features were not always available as epilepsy age at onset and type of seizure. Lastly, we cannot exclude reverse causality because there are no prospective population-based survey including new cases in the literature.

Public health implications

Human toxocariasis is a cosmopolitan and preventable infection that may have a role in the global burden of epilepsy. There are however many available strategies to reduce the impact of this neglected diseases: public health policies could limit soil contamination and control *Toxocara* infection in definitive hosts [58] and enhancing education will improve the public's understanding of toxocariasis and its prevention [5]. As an alternative, interest has grown recently to develop vaccines against helminth infections, specifically to prevent transmission of zoonotic diseases to human beings [59].

Conclusion

We provide epidemiological evidence of a positive association between epilepsy and toxocariasis seroprevalence. New surveys supported the association, mainly population-based studies. We showed that research interest is slightly increasing in the field and studies with appropriate methodological approach have been produced in the last years. We also highlighted the presence of several knowledge gaps that have not been addressed. We strongly encourage researchers to perform further studies in the field to answer these questions. Further research should be performed; ideally, including new cases in a general population settings, to better explore the association or assessing the reduction in the incidence of epilepsy following the eradication of Toxocariasis. Increasing the understanding of physiopathological mechanisms of epileptogenesis will clarify the role of certain parasites in epilepsy.

Supporting information

S1 File. PRISMA checklist. (DOC)

S1 Fig. PRISMA flowchart. (EPS)

S2 Fig. Funnel plot. (EPS)

Author Contributions

Conceptualization: Graziella Quattrocchi, Alessandra Nicoletti, Pierre-Marie Preux.

Data curation: Jaime Luna, Calogero Edoardo Cicero, Guillaume Rateau, Graziella Quattrocchi.

Formal analysis: Jaime Luna, Guillaume Rateau.

Investigation: Jaime Luna, Calogero Edoardo Cicero, Guillaume Rateau, Graziella Quattrocchi.

Methodology: Jaime Luna, Graziella Quattrocchi, Alessandra Nicoletti, Pierre-Marie Preux.

Software: François Dalmay.

Supervision: Alessandra Nicoletti, Pierre-Marie Preux.

Validation: Alessandra Nicoletti, Pierre-Marie Preux.

Writing - original draft: Jaime Luna, Calogero Edoardo Cicero.

Writing – review & editing: Jaime Luna, Calogero Edoardo Cicero, Guillaume Rateau, Graziella Quattrocchi, Benoit Marin, Elisa Bruno, François Dalmay, Michel Druet-Cabanac, Alessandra Nicoletti, Pierre-Marie Preux.

References

- Fan C-K, Holland CV, Loxton K, Barghouth U. Cerebral Toxocariasis: Silent Progression to Neurodegenerative Disorders? Clin Microbiol Rev. 2015 Jul; 28(3):663–86. https://doi.org/10.1128/CMR.00106-14 PMID: 26062575
- Glickman LT, Schantz PM. Epidemiology and pathogenesis of zoonotic toxocariasis. Epidemiol Rev. 1981; 3:230–50. PMID: 7030762
- Schantz PM, Glickman LT. Toxocaral visceral larva migrans. N Engl J Med. 1978 Feb 23; 298(8):436– 9. https://doi.org/10.1056/NEJM197802232980806 PMID: 622118
- 4. Deshayes S, Bonhomme J, de La Blanchardière A. Neurotoxocariasis: a systematic literature review. Infection. 2016 Oct; 44(5):565–74. https://doi.org/10.1007/s15010-016-0889-8 PMID: 27084369
- Ma G, Holland CV, Wang T, Hofmann A, Fan C-K, Maizels RM, et al. Human toxocariasis. Lancet Infect Dis. 2018 Jan; 18(1):e14–24. https://doi.org/10.1016/S1473-3099(17)30331-6 PMID: 28781085
- Quattrocchi G, Nicoletti A, Marin B, Bruno E, Druet-Cabanac M, Preux P-M. Toxocariasis and epilepsy: systematic review and meta-analysis. PLoS Negl Trop Dis. 2012; 6(8):e1775. https://doi.org/10.1371/ journal.pntd.0001775 PMID: 22905274

- Rubinsky-Elefant G, Hirata CE, Yamamoto JH, Ferreira MU. Human toxocariasis: diagnosis, worldwide seroprevalences and clinical expression of the systemic and ocular forms. Ann Trop Med Parasitol. 2010 Jan; 104(1):3–23. https://doi.org/10.1179/136485910X12607012373957 PMID: 20149289
- Salem G, Schantz P. Toxocaral visceral larva migrans after ingestion of raw lamb liver. Clin Infect Dis Off Publ Infect Dis Soc Am. 1992 Oct; 15(4):743–4.
- Yoshikawa M, Nishiofuku M, Moriya K, Ouji Y, Ishizaka S, Kasahara K, et al. A familial case of visceral toxocariasis due to consumption of raw bovine liver. Parasitol Int. 2008 Dec; 57(4):525–9. <u>https://doi.org/10.1016/j.parint.2008.08.002</u> PMID: 18771749
- Rojas TO, Romero C, Heredia R, Bautista LG, Sheinberg G. Identification of Toxocaraspp. eggs in dog hair and associated risk factors. Vet World. 2017 Jul; 10(7):798–802. https://doi.org/10.14202/vetworld. 2017.798-802 PMID: 28831225
- Overgaauw PAM, van Zutphen L, Hoek D, Yaya FO, Roelfsema J, Pinelli E, et al. Zoonotic parasites in fecal samples and fur from dogs and cats in The Netherlands. Vet Parasitol. 2009 Jul 7; 163(1–2):115– 22. https://doi.org/10.1016/j.vetpar.2009.03.044 PMID: 19398275
- Pawlowski Z. Toxocariasis in humans: clinical expression and treatment dilemma. J Helminthol. 2001 Dec; 75(4):299–305. PMID: <u>11818044</u>
- Macpherson CNL. The epidemiology and public health importance of toxocariasis: a zoonosis of global importance. Int J Parasitol. 2013 Nov; 43(12–13):999–1008. <u>https://doi.org/10.1016/j.ijpara.2013.07</u>. 004 PMID: 23954435
- 14. Nicoletti A. Toxocariasis. Handb Clin Neurol. 2013; 114:217–28. <u>https://doi.org/10.1016/B978-0-444-53490-3.00016-9</u> PMID: 23829912
- Sánchez SS, García HH, Nicoletti A. Clinical and Magnetic Resonance Imaging Findings of Neurotoxocariasis. Front Neurol. 2018; 9:53. https://doi.org/10.3389/fneur.2018.00053 PMID: 29472889
- 16. Beautyman W, Woolf AL. An ascaris larva in the brain in association with acute anterior poliomyelitis. J Pathol Bacteriol. 1951 Oct; 63(4):635–47. PMID: 14898369
- Graeff-Teixeira C, da Silva ACA, Yoshimura K. Update on eosinophilic meningoencephalitis and its clinical relevance. Clin Microbiol Rev. 2009 Apr; 22(2):322–348, Table of Contents. <u>https://doi.org/10.1128/</u> CMR.00044-08 PMID: 19366917
- Woodruff AW, Bisseru B, Bowe JC. Infection with animal helminths as a factor in causing poliomyelitis and epilepsy. Br Med J. 1966 Jun 25; 1(5503):1576–9. PMID: 5949491
- Critchley EM, Vakil SD, Hutchinson DN, Taylor P. Toxoplasma, Toxocara, and epilepsy. Epilepsia. 1982 Jun; 23(3):315–21. PMID: 7084141
- 20. Epi Info «Epi Info 6» package (1994) World Health Organization, Geneva, Switzerland, Centers for Disease Control & Prevention (CDC), Atlanta, GA.
- Borenstein M, Hedges LV, Higgins JPT, Rothstein HR. A basic introduction to fixed-effect and randomeffects models for meta-analysis. Res Synth Methods. 2010 Apr; 1(2):97–111. <u>https://doi.org/10.1002/</u> jrsm.12 PMID: 26061376
- Thompson SG, Sharp SJ. Explaining heterogeneity in meta-analysis: a comparison of methods. Stat Med. 1999 Oct 30; 18(20):2693–708. PMID: 10521860
- Nicoletti A, Bartoloni A, Reggio A, Bartalesi F, Roselli M, Sofia V, et al. Epilepsy, cysticercosis, and toxocariasis: a population-based case-control study in rural Bolivia. Neurology. 2002 Apr 23; 58(8):1256– 61. PMID: 11971095
- Nicoletti A, Bartoloni A, Sofia V, Mantella A, Nsengiyumva G, Frescaline G, et al. Epilepsy and toxocariasis: a case-control study in Burundi. Epilepsia. 2007 May; 48(5):894–9. https://doi.org/10.1111/j.1528-1167.2007.01104.x PMID: 17508999
- Nicoletti A, Sofia V, Mantella A, Vitale G, Contrafatto D, Sorbello V, et al. Epilepsy and toxocariasis: a case-control study in Italy. Epilepsia. 2008 Apr; 49(4):594–9. <u>https://doi.org/10.1111/j.1528-1167.2007.01432.x PMID: 18031545</u>
- Winkler AS, Blocher J, Auer H, Gotwald T, Matuja W, Schmutzhard E. Anticysticercal and antitoxocaral antibodies in people with epilepsy in rural Tanzania. Trans R Soc Trop Med Hyg. 2008 Oct; 102 (10):1032–8. https://doi.org/10.1016/j.trstmh.2008.05.004 PMID: 18571212
- 27. Singh G, Bawa J, Chinna D, Chaudhary A, Saggar K, Modi M, et al. Association between epilepsy and cysticercosis and toxocariasis: a population-based case-control study in a slum in India. Epilepsia. 2012 Dec; 53(12):2203–8. https://doi.org/10.1111/epi.12005 PMID: 23106145
- Magnaval JF, Fabre R, Maurières P, Charlet JP, de Larrard B. Application of the western blotting procedure for the immunodiagnosis of human toxocariasis. Parasitol Res. 1991; 77(8):697–702. PMID: 1805214

- Glickman LT, Cypess RH, Crumrine PK, Gitlin DA. Toxocara infection and epilepsy in children. J Pediatr. 1979 Jan; 94(1):75–8. PMID: 758428
- Arpino C, Gattinara GC, Piergili D, Curatolo P. Toxocara infection and epilepsy in children: a case-control study. Epilepsia. 1990 Feb; 31(1):33–6. PMID: 2303010
- El-Tantawy NL, El-Nahas HA, Salem DA-B, Salem NA-B, Hasaneen BM. Seroprevalence of *Toxo-plasma Gondii* and *Toxocara Spp* in Children with Cryptogenic Epilepsy. Am J Infect Dis Microbiol Am J Infect Dis Microbiol. 2013 Jan 23; 1(5):92–5.
- 32. Kamuyu G, Bottomley C, Mageto J, Lowe B, Wilkins PP, Noh JC, et al. Exposure to multiple parasites is associated with the prevalence of active convulsive epilepsy in sub-Saharan Africa. PLoS Negl Trop Dis. 2014; 8(5):e2908. https://doi.org/10.1371/journal.pntd.0002908 PMID: 24875312
- Gail M. Case–Control Study, Population-Based—Encyclopedia of Biostatistics. (2nd ed. Hoboken, NJ: Wiley, 2005).
- Moher D, Liberati A, Tetzlaff J, Altman DG, PRISMA Group. Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement. PLoS Med. 2009 Jul 21; 6(7):e1000097. <u>https:// doi.org/10.1371/journal.pmed.1000097</u> PMID: 19621072
- Zibaei M, Firoozeh F, Bahrami P, Sadjjadi SM. Investigation of anti-Toxocara antibodies in epileptic patients and comparison of two methods: ELISA and Western blotting. Epilepsy Res Treat. 2013; 2013:156815. https://doi.org/10.1155/2013/156815 PMID: 23710354
- Nkouawa A, Sako Y, Itoh S, Kouojip-Mabou A, Nganou CN, Saijo Y, et al. Serological studies of neurologic helminthic infections in rural areas of southwest cameroon: toxocariasis, cysticercosis and paragonimiasis. PLoS Negl Trop Dis. 2010 Jul 6; 4(7):e732. <u>https://doi.org/10.1371/journal.pntd.0000732</u> PMID: 20625553
- Eraky MA, Abdel-Hady S, Abdallah KF. Seropositivity of Toxoplasma gondii and Toxocara spp. in Children with Cryptogenic Epilepsy, Benha, Egypt. Korean J Parasitol. 2016 Jun; 54(3):335–8. https://doi.org/10.3347/kjp.2016.54.3.335 PMID: 27417090
- Ngugi AK, Bottomley C, Kleinschmidt I, Wagner RG, Kakooza-Mwesige A, Ae-Ngibise K, et al. Prevalence of active convulsive epilepsy in sub-Saharan Africa and associated risk factors: cross-sectional and case-control studies. Lancet Neurol. 2013 Mar; 12(3):253–63. https://doi.org/10.1016/S1474-4422 (13)70003-6 PMID: 23375964
- Akyol A, Bicerol B, Ertug S, Ertabaklar H, Kiylioglu N. Epilepsy and seropositivity rates of Toxocara canis and Toxoplasma gondii. Seizure. 2007 Apr; 16(3):233–7. <u>https://doi.org/10.1016/j.seizure.2006</u>. 12.008 PMID: 17239629
- Guidelines for epidemiologic studies on epilepsy. Commission on Epidemiology and Prognosis, International League Against Epilepsy. Epilepsia. 1993 Aug; 34(4):592–6. PMID: 8330566
- Edwards T, Scott AG, Munyoki G, Odera VM, Chengo E, Bauni E, et al. Active convulsive epilepsy in a rural district of Kenya: a study of prevalence and possible risk factors. Lancet Neurol. 2008 Jan; 7 (1):50–6. https://doi.org/10.1016/S1474-4422(07)70292-2 PMID: 18068520
- Alter M, Allen H. Proposed definitions and classifications of epilepsy for epidemiological purposes. Epidemiolgy Epilepsy Workshop US Dept HEW NIH NINDS 147–148.
- Senanayake N, Román GC. Epidemiology of epilepsy in developing countries. Bull World Health Organ. 1993; 71(2):247–58. PMID: 8490989
- Proposal for revised clinical and electroencephalographic classification of epileptic seizures. From the Commission on Classification and Terminology of the International League Against Epilepsy. Epilepsia. 1981 Aug; 22(4):489–501. PMID: 6790275
- 45. Winkler AS, Schaffert M, Schmutzhard E. Epilepsy in resource poor countries—suggestion of an adjusted classification. Epilepsia. 2007 May; 48(5):1029–30. <u>https://doi.org/10.1111/j.1528-1167.2007</u>. 01009_1.x PMID: 17509005
- Young JM, Solomon MJ. How to critically appraise an article. Nat Clin Pract Gastroenterol Hepatol. 2009 Feb; 6(2):82–91. https://doi.org/10.1038/ncpgasthep1331 PMID: 19153565
- Wacholder S, McLaughlin JK, Silverman DT, Mandel JS. Selection of Controls in Case-Control StudiesI. Principles. Am J Epidemiol. 1992 May 1; 135(9):1019–28. PMID: <u>1595688</u>
- **48.** Oteifa NM, Moustafa MA. The potential risk of contracting toxocariasis in Heliopolis district, Cairo, Egypt. J Egypt Soc Parasitol. 1997 Apr; 27(1):197–203. PMID: <u>9097541</u>
- **49.** de Savigny DH, Voller A, Woodruff AW. Toxocariasis: serological diagnosis by enzyme immunoassay. J Clin Pathol. 1979 Mar; 32(3):284–8. PMID: 372253
- Fillaux J, Magnaval J-F. Laboratory diagnosis of human toxocariasis. Vet Parasitol. 2013 Apr 15; 193 (4):327–36. https://doi.org/10.1016/j.vetpar.2012.12.028 PMID: 23318165

- **51.** Wagner RG, Newton CR. Do helminths cause epilepsy? Parasite Immunol. 2009 Nov; 31(11):697–705. https://doi.org/10.1111/j.1365-3024.2009.01128.x PMID: 19825109
- Nash TE, Mahanty S, Loeb JA, Theodore WH, Friedman A, Sander JW, et al. Neurocysticercosis: A natural human model of epileptogenesis. Epilepsia. 2015 Feb; 56(2):177–83. <u>https://doi.org/10.1111/</u> epi.12849 PMID: 25534640
- 53. Levite M. Autoimmune epilepsy. Nat Immunol. 2002 Jun; 3(6):500. https://doi.org/10.1038/ni0602-500 PMID: 12032558
- Ganor Y, Goldberg-Stern H, Lerman-Sagie T, Teichberg VI, Levite M. Autoimmune epilepsy: distinct subpopulations of epilepsy patients harbor serum autoantibodies to either glutamate/AMPA receptor GluR3, glutamate/NMDA receptor subunit NR2A or double-stranded DNA. Epilepsy Res. 2005 Jun; 65 (1–2):11–22. https://doi.org/10.1016/j.eplepsyres.2005.03.011 PMID: 15978777
- 55. Palace J, Lang B. Epilepsy: an autoimmune disease? J Neurol Neurosurg Psychiatry. 2000 Dec; 69 (6):711–4. https://doi.org/10.1136/jnnp.69.6.711 PMID: 11080217
- 56. Levite M, Ganor Y. Autoantibodies to glutamate receptors can damage the brain in epilepsy, systemic lupus erythematosus and encephalitis. Expert Rev Neurother. 2008 Jul; 8(7):1141–60. https://doi.org/ 10.1586/14737175.8.7.1141 PMID: 18590483
- Obwaller A, Duchêne M, Walochnik J, Wiedermann G, Auer H, Aspöck H. Association of autoantibodies against small nuclear ribonucleoproteins (snRNPs) with symptomatic Toxocara canis infestation. Parasite Immunol. 2004 Sep; 26(8–9):327–33. <u>https://doi.org/10.1111/j.0141-9838.2004.00716.x</u> PMID: 15679629
- Moreira GMSG, Telmo P de L, Mendonça M, Moreira AN, McBride AJA, Scaini CJ, et al. Human toxocariasis: current advances in diagnostics, treatment, and interventions. Trends Parasitol. 2014 Sep; 30 (9):456–64. https://doi.org/10.1016/j.pt.2014.07.003 PMID: 25089038
- Garcia HH, Gonzalez AE, Tsang VCW, O'Neal SE, Llanos-Zavalaga F, Gonzalvez G, et al. Elimination of Taenia solium Transmission in Northern Peru. N Engl J Med. 2016 Jun 16; 374(24):2335–44. <u>https:// doi.org/10.1056/NEJMoa1515520 PMID: 27305193</u>